Author’s response to reviews

Title: Biopsy-proven vancomycin-induced acute kidney injury: a case report and literature review

Authors:
ANRI SAWADA (anri-sawada@nms.ac.jp)
Kunio Kawanishi (kukawanishi@ucsd.edu)
Shohei Morikawa (morikawa.shohei@kameda.jp)
Toshihiro Nakano (t.nakano.dr@gmail.com)
Mio Kodama (mio.dama1113@hotmail.co.jp)
Mitihiro Mitobe (mitobe.michihiro@kameda.jp)
Sekiko Taneda (sekikosekiko@gmail.com)
Junki Koike (j2koike@marianna-u.ac.jp)
Mamiko Ohara (oharam-tyk@umin.net)
Yoji Nagashima (nagashima.yoji@twmu.ac.jp)
Kosaku Nitta (knitta@twmu.ac.jp)
Takahiro Mochizuki (mochizuki.takahiro@kameda.jp)

Version: 3 Date: 28 Dec 2017

Author’s response to reviews:

Author’s response to reviews

Title: Biopsy-proven vancomycin-induced acute kidney injury: a case report and literature review

Authors:
Anri Sawada (anri-sawada@nms.ac.jp)
Dear Dr. Steven Weisbord, Editor of BMC series

We appreciate meaningful comments from reviewers, and your editors for our manuscript entitled "Biopsy-proven vancomycin-induced acute kidney injury: a case report and literature review" (Manuscript number BNEP-D-17-00525R2). Herein, I am going to send a revised manuscript that would hopefully be responded to all comments by you and reviewers. Each correction in manuscript was highlighted in red color.

Thank you for your consideration.
Sincerely yours,

Anri Sawada, Department of Surgical Pathology, Tokyo Women's Medical University, 8-1 Kawadacho, Shinjuku, Tokyo, 162-8666, Japan
E-mail : anri-sawada@nms.ac.jp, Tel : +81-3-3353-8111

Response to M. Barry Stokes (Reviewer 1)

We wish to express our deep appreciation to you to evaluate our manuscript (Manuscript number BNEP-D-17-00525R2).

Thank you for approving our revised manuscript.

No comments to address.

Response to Ying Zhou (Reviewer 2)

We wish to express our deep appreciation to you to evaluate our manuscript (Manuscript number BNEP-D-17-00525R2). Our responses to your comments and revised manuscript are as follows:

Comment 1:
The white blood cell count on P6 L51 seems incorrect, which should be 25700/μL, corresponding to Table 1 (P26).

Response1:

According to your suggestion, we corrected blood cell count in page 6, line 15-16.

(versions)

Inflammatory markers were elevated (white blood cell count 25700 cells/m3 with left shift and C reactive protein 28.8 mg/L).

(versions)

Inflammatory markers were elevated (white blood cell count 25700/μL with left shift and C reactive protein 28.8 mg/L).